AN OVERVIEW OF CORPORATE GOVERNANCE PRACTICES: AN INSTITUTIONAL PERSPECTIVE IN AN EMERGING MARKET

Jebreel Mohammad Al-Msiedeen^{*}, Hashem Alshurafat^{**}, Mohammad K. Shbeilat^{***}, Abdullah Majed Al Maani^{****}, Lubna Khaled Alroud^{*}

* Faculty of Business, Tafila Technical University, Tafila, Jordan
** Faculty of Economics and Administrative Sciences, The Hashemite University, Zarqa, Jordan
*** Corresponding author, College of Business, Tafila Technical University, Tafila, Jordan
Contact details: College of Business, Tafila Technical University, P. O. Box 179, Tafila 66110, Jordan
**** Faculty of Business and Economics, Al-Hussein Bin Talal University, Maan, Jordan



How to cite this paper: Al-Msiedeen, J. M., Alshurafat, H., Shbeilat, M. K., & Al Maani, A. M., & Alroud, L. K. (2024). An overview of corporate governance practices: An institutional perspective in an emerging market. *Corporate Governance and Organizational Behavior Review*, 8(1), 191–204. https://doi.org/10.22495/cgobrv8ilp16

Copyright © 2024 The Authors

This work is licensed under a Creative Commons Attribution 4.0 International License (CC BY 4.0). https://creativecommons.org/licenses/by/ 4.0/

ISSN Online: 2521-1889 ISSN Print: 2521-1870

Received: 21.08.2023 Accepted: 12.02.2024

JEL Classification: G38, M48, O16 **DOI:** 10.22495/cgobrv8ilp16

Abstract

This study provides a comprehensive view of the current practice of corporate governance (CG) in the emerging market, with a specific focus on Jordan. By using archival data, namely the academic literature, reports published by the main institutions, and financial reports published by listed companies, this study addresses the roles explored of main institutional forces, firms' characteristics, financial markets, and the Jordanian economic environment in reinforcing the current practices of CG in Jordan. Although Jordan aims for institutional development to reinforce practices toward consistency with international CG best practices, the results of this study show that there is a weak institutional environment, with a seeming lack of some external control instruments. The findings also indicate that unpredictable legal enforcement causes poor compliance by Jordanian companies. Therefore, current CG practices in Jordan are neither adequate nor comprehensive (Mansour et al., 2023). Therefore, the weakness of CG may lead to further financial crises at the company level in Jordan. This study represents the first attempt to explore whether institutional forces are essential to strengthen CG practices in developing nations, and the results of this article can serve as a prototype for other developing economies.

Keywords: Jordan, New Institutional Sociology, Corporate Governance

Authors' individual contribution: Conceptualization — J.M.A.-M., H.A., and M.K.S.; Methodology — J.M.A.-M., A.M.A.M., and L.K.A.; Formal Analysis — H.A. and A.M.A.M.; Investigation — J.M.A.-M., A.M.A.M., and L.K.A.; Resources — J.M.A.-M., M.K.S., and A.M.A.M.; Writing — Original Draft — J.M.A.-M. and H.A.; Writing — Review & Editing — J.M.A.-M. and M.K.S.; Visualization — H.A. and A.M.A.M.; Supervision — J.M.A.-M. and H.A.; Project Administration — J.M.A.-M. and M.K.S.

Declaration of conflicting interests: The Authors declare that there is no conflict of interest.

1. INTRODUCTION

Corporate governance (CG) is a longstanding yet relevant issue in accounting fields and academic

writing due to many corporate collapses around the world, such as those during the Asian and Global Financial Crises. CG practice monitors management's effectiveness, protecting shareholders' rights and

VIRTUS 191

mitigating noncompliance incidences by preventing egregious behavior. In this vein, leading international bodies such as the World Bank, the International Monetary Fund (IMF), the International Finance Corporation (IFC), and the Organization for Economic Cooperation and Development (OECD) strongly promote the development of guidelines, regulations, standards and codes of governance around the world. Epps and Cereola (2008) noted that the OECD described the CG good practice as having "implications for company behavior toward employees, shareholders, customers and banks" (p. 1136). The World Bank contends that governance aids nations in constructing efficient, capable, accountable and inclusive institutions. Those with producing healthy institutions by progress an environment which decreases poverty and provides valuable services (World Bank, 2023). In developing countries, compliance with good practices of CG is an effective tool when regulatory and legal environments are weak, with such environments typified by a predominantly high concentration of ownership.

In this context, firms that encourage good CG practices could protect their owners' interests and improve their investors' confidence (La Porta et al., 2000). Earlier researchers have recognized the features of the institutional framework in developing countries (Aguilera & Jackson, 2010; Siddiqui, 2010; Rashid, 2011; Iswaissi & Falahati, 2017). Davis (2005) argued that new institutional sociological theory offers a comprehensive understanding of the formal and informal factors, values, norms, and culture of the company in modern society. Likewise, agency theory points to a conflict of interest between management and shareholders due to the separation of ownership and control. To alleviate the firm's agency cost, agency perspectives suggest firms should adopt internal and external CG mechanisms (Al-Najjar & Clark, 2017). In this article, the institutional theory is adopted as the framework for understanding the institutional environment's weaknesses in the Jordanian corporate sector.

Jordan is characterized by poor CG practice, weak institutional framework and highly а concentrated ownership. In Jordan, financial scandals have occurred, including the 1989 bankruptcy of Petra Bank and the drop of Jordan's dinar from USD3.35 to USD1.41 in the same year (Shbeilat & Abdel-Qader, 2018). Following this, Jordan has witnessed 44 bankruptcies of Jordanian companies. As a consequence, its economy faces many challenges: a budget deficit, high government debt, high unemployment and poverty, and waves of COVID-19 infections, leading to continued economic deterioration (World Bank, 2021), causing more substantial debate about CG in Jordan. This article offers new evidence on existing CG practices in Jordan's institutional context, about which little is known.

The motivation for this research is to seek improvements in the legal, institutional, and regulatory environment to promote transparency and accountability in the less-researched Jordanian corporate sector. This is done by exploring the factors affecting improved good CG practices and, ultimately, increasing investor confidence in Jordan. This article explores the roles of main institutional forces, firms' characteristics, financial markets, and the Jordanian economic environment in shaping existing CG practices in Jordan. The research questions are:

RQ1: Are institutional regulatory bodies exercising sufficient pressure on companies so that they reliably follow corporate governance principles and regulations?

RQ2: Do Jordanian companies adhere to the authoritative guidelines of good corporate governance practice?

RQ3: What obstacles hinder the implementation of corporate governance good practices in the Jordanian corporate sector?

Archival data is used for this article: academic literature, reports published by the main institutions, and financial reports published by listed companies.

The remainder of this research is organized as follows. Section 2 reviews the literature. Section 3 describes the methodology. Section 4 presents the theoretical framework. Section 5 provides results and discusses the professions of accounting and auditing in Jordan. Section 6 concludes the study.

2. LITERATURE REVIEW

Corporate governance mechanisms have emerged among the most important issues to address in both developed and developing countries. Globally, corporate financial scandals, failures, and collapses have prompted concern in Jordan's government which has considered various procedures to protect and augment the nation's financial environment.

2.1. Jordanian corporate governance: An overview

2.1.1. Regulators and governance

Corporate governance practices are relatively new in Jordan. In 1978, the Amman Financial Market (AFM) was established to regulate companies. In the 1990s, due to international organizations' increased interest in CG practices in Jordan (e.g., the OECD, IFC and World Bank), numerous CG reforms were initiated with their support. In 1997, the AFM was replaced by three official institutions, (i.e., the Jordan Securities Commission — JSC¹, the Amman Stock Exchange — ASE², and the Securities Depository Center — SDC³).

In the same year, Central Bank of Jordan (CBJ) Law No. 23 (CBJ, 1971) was an important landmark for the Jordanian capital market. Consequently, the number of firms listed on the ASE grew gradually from 1998. especially till 2011. Nevertheless. the Jordan stock market experienced major turmoil in 2000, with 44 bankruptcy cases during this period: 26 in the industrial sector, 15 in services, and three in finance (Zureigat et al., 2014). Some have attributed this to weak CG practices. As a result, international organizations proposed several improvements to governance practices. Consequently, CG has been increasingly debated in Jordan (Al-Msiedeen, 2019), with many CG codes issued. These comprise CG instructions for the shareholding of non-listed public shareholding, listed, and limited liability companies, as well as for private shareholding firms, banks and insurance companies.

The OECD's CG principles emerged as the international benchmark for good CG practice,

¹ https://jsc.gov.jo/Default/en ² https://www.ase.com.jo/en

² https://www.ase.com.jo/en ³ https://www.sdc.com.jo/english/

VIRTUS

with widespread adoption. These principles comprise the foundations of an effective CG framework, the main ownership functions, shareholders' rights and equitable treatment, stakeholders' role in CG, the board's responsibilities, transparency, and disclosure. Jordan adopted a series of CG codes with these principles through three main steps, as explained below. Later updates required all Jordanian-listed companies to include a governance report in their annual reports. This code included definitions for independent directors, the number of board members, board independence, chief executive officer (CEO) duality, stakeholders' rights, disclosure, and transparency.

2.1.2. Jordanian codes of corporate governance

Organization for Economic Cooperation and Development principles have been essential in strengthening CG codes internationally, including in Jordan. They proscribe a constructive CG framework, major ownership functions, shareholders' rights and what constitutes their equitable treatment, stakeholders' role in CG, transparency, disclosure and board responsibilities. These principles indicate what constitutes good CG practice for regulators, participants policymakers, and market for supporting the regulatory, legal, and institutional framework underpinning CG. They give practical instructions for firms, stock exchanges, investors, and other parties involved in developing good CG practices. Thus, these principles have taken center stage in the construction of an environment of transparency, accountability, trust, and the requisite business integrity and financial stability which promote these elements, thereby supporting more robust economic growth.

In 2005, the JSC issued a non-mandatory code for ASE-listed companies. In 2009, a revised 'comply or explain' approach to the code was established. In 2017, Jordan's government issued new CG instructions for ASE-listed firms, applying international standards. Based on the Insurance Regulatory Law No. 33 of 1999 and its amendments, the Insurance Commission (IC) issued CG instructions specific to insurance companies in Jordan. For the banking sector, various steps were taken: the 2004 bank directors' CG handbook, the 2007 CG code for banks, and the 2016 amended CG bank instructions. These also followed the Basel Committee's guidance for banking supervision. These spoke to various aspects of CG, including the board of directors' responsibilities and experience. the board's independence, supervision mechanisms, and the bank's internal control system.

2.2. Mechanisms of corporate governance in Jordan

In Jordan, internal factors, such as its characteristically weak institutional forces, the Anglo-American model is not suitable. Similarly, to Asia countries, such as Japan, and many European countries, such as the Netherlands and Germany, economic activities in the Jordanian stock market are dependent on mutual or personal relations. The mechanisms of corporate control in Jordan are often insideroriented (e.g., due to the ownership structure) (Alkurdi & Mardini, 2020). The main investors have a stake of significant ownership within Jordanian firms and, in general, are board members. In other words, the ownership of a company's shares is highly concentrated in the hands of a small number of investors who have a pivotal role in disciplining the companies through close relationships. Therefore, the market is not a tool for firm control in the Jordanian corporate sector (Alshirah et al., 2022). It may thus be argued that Jordan has a high level of ownership concentration, a lack of takeover regulations and an inefficient market.

With 44 bankruptcy cases in Jordanian companies from 2000 to 2011 (Zureigat et al., 2014), Jordan is struggling toward comprehensive institutional development to strengthen CG practices consistent with international CG best practices. The establishment of the ASE, the JSC, the SDC, and the Jordanian Institute of Directors, formed a turning point for the Jordanian market. These reforms contributed to the development of a regulatory and legal environment for sustained economic growth.

2.2.1. Corporate ownership structure

The separation of ownership structure from management is rare. Unlike Anglo-American firms, ownership concentration in Jordan is intensely high and cross-shareholding is not very popular, which means that individual shareholdings are very large. Thus, joint stock firms in Jordan are heavily controlled by the dominant shareholders (founding sponsors/directors). In general, these are family members with positions on the board of directors. Therefore, the Jordanian corporate governance (JCG) model may be termed an "ownership-based model" (Al-Msiedeen & Al Sawalqa, 2021; Al-Begali & Phua, 2023). From a sample of 118 listed nonfinancial companies in Jordan, Kanakrivah (2021) indicated that the average proportion of shares owned by members of the board of directors is 71.8%. In a few companies, ownership is controlled by the government and foreign investors. Hence, most decision-making in a Jordanian company is done by a few managers who are representatives of the family owning the company. This makes it difficult for non-controlling shareholders to exercise their ownership rights, such as by removing poorly performing management or appointing managers in the first place.

2.2.2. Board of directors

Jordan's Companies Law No 22 of 1997 and its amendments provide guidelines about the appointment, roles, and responsibilities of members of the board. However, there is no clear guideline on board leadership structure and board independence. This law and its amendments function in opposition to the more recent 2017 CG instructions for shareholding listed companies. For instance, the Companies Law No 22 of 1997 requirement that all board members must be shareholders works against the concept of directors' independence. Section 133(C) of the Jordanian Companies Law No 22 of 1997 and its amendments states that the membership of any member of the board of directors of a firm listed on the ASE shall be forfeited if the number of shares owned by the member is less than the agreed limit. Therefore, the appointment of professional directors to boards is curtailed by this law. Moreover,



the requirement that board members be shareholders appears inconsistent with the concept of an independent director.

The Companies Law No 22 of 1997 and its amendments make no mention of the experience and qualifications of an independent director as directors are not considered to be independent. As well, there is no strict condition for appointing board members when a position becomes vacant; such a member is elected by the other board members. Given the status quo in Jordan, explained above, it is likely that the process of appointment is usually based on a close personal relationship with the other board members, large shareholders, and firm managers, rather than any experience and qualifications for the role. Al-Msiedeen et al.'s (2018) empirical research on board independence (as represented by external independent directors) reveals that, overall, board independence positively influences firm performance in Jordan. Altawalbeh (2020) points out that only 52% of nonfinancial firms listed on the ASE have at least one independent director. This supports the claim by the European Bank for Reconstructions and Development that few independent directors of the largest Jordanian-listed companies are disclosed (Cigna & Sigheartau, 2017). It follows that many Jordanian companies have not appointed any independent directors at all.

Anglo-American Similarly, to countries. boardrooms in Jordan are organized under a onetier system, while firms in many European countries, such as Germany and the Netherlands, are organized as two tiers. Under the former, the CEO and board chairperson carry out their responsibilities and duties together (Maassen, 1999). In the two-tier system, the board's executive function is separate from its monitoring function (Maassen, 1999). In practice, Jordanian boardrooms are strongly controlled by owners who belong to one family. Therefore, family duality is prevalent: the father serves as board chairman and the son serves as CEO.

2.2.3. Management and chief executive officer

The term "CEO" is not commonly used in Jordan. The new code of CG directives for firms listed on the ASE and the Jordanian Companies Law No 22 of 1997 (and its amendments) provide some guidelines concerning the position of the director who is the CEO in fact (see Section 152(C) of the Companies Law No 22 of 1997 and its amendments). In most Jordanian firms, the CEO is the representative of family members or the majority of shareholders. More than half of Jordanian companies are owned and controlled by family members (Salameh et al., 2023), which can lead to a lack of focus on their expertise and qualifications in their appointment. Hence, this situation may lead to lower accountability of management to the board members.

As firms in Jordan operate under a one-tier board system, the non-executive and executive managers hold the positions of board chairman and CEO, respectively, and tend to be closely related, if not by blood. However, as noted by Abdullatif et al. (2019), the concept of CEO duality is currently very limited in the Jordanian corporate sector, especially in banking and insurance. Consequently, it seems that JCG needs many reforms, especially in terms of the value placed on an independent board member.

VIRTUS

The present situation gives substantial power, authority, and decision-making to CEOs, therefore, mitigating the ability of the board to exercise governance (by monitoring) and leading to an agency cost by creating a conflict of interest between the board members and management.

Such a conflict of interest may diminish the board's independence and therefore create severe problems, as shown by Enron and other spectacular corporate collapses with the same cause. As noted by Cornett et al. (2007), "CEO/chair duality concentrates power with the CEO, potentially making disagreement on the part of outsiders costlier, which can exacerbate potential conflicts of interest" (p. 1775).

2.2.4. Shareholders rights

Generally speaking, the Jordanian Companies Law No 22 of 1997 and its amendments and the Jordanian Corporate Governance Code (JCGC) 2017 (JSC, 2017) provide several main rights to shareholders (e.g., rights of appointments to the board of directors, right of dividend, and preemptive rights to subscribe additional shares). Such rights can be exercised by shareholders through voting for directors and on other main issues in the shareholders' meeting. The Jordanian Companies Law No 22 of 1997 and its amendments provide many provisions for such rights. Based on equal treatment of shareholders, for instance, Section 98(D) requires that all shareholders of a firm have the right to access the shareholders register in the company regarding their ownership, for whatever reason. Further, Section 157(A) indicates that the chairperson and members of a public shareholding firm board shall be held responsible to the shareholders, firm, and others for every violation committed, by any or all of the members, of the regulations and laws in force, of the firm's memorandum of association, and for any error in the company's management. JCGC 2017, Section 13(5) proscribes the participation and voting in the general assembly meeting of a firm, whether in person or by proxy, with votes equal to the number of stocks owned by the shareholder of the firm and delegated stocks. To protect shareholders from dilution, Section 13(7) of JCGC 2017 states the right of preemptive to any new company issues prior to displaying them to other investors, unless the shareholders of the company waive this priority by a decision through an extraordinary general assembly of the firm (JSC, 2017).

Rashid (2011) confirms that "there is no legislative guideline for calling a 'meeting of shareholders' by shareholders" (p. 20) in many developed countries. However, in a less developed country like Jordan, shareholders holding not less than 10% of the shares of the company can request extraordinary general assembly meeting, an mentioned in Section 13(10) of JCGC 2017. as Section 13(11) of JCGC 2017 points out as well that shareholders holding not less than 20% of the shares of the company can request an extraordinary general assembly meeting to seek the dismissal of the chairperson or any member of the board of director, excluding members representing any public legal person and government shares. Further, only shareholders holding not less than 10% of the shares of the company can request an audit of the company's records and its activities.

In fact, many company shareholders are not aware of their main rights and rarely have the motivation to exercise these rights. It may also be argued that many shareholders are not given adequate opportunities to participate in decisions related to the company (Pearce, 2015). They may not have enough skills to exert pressure on the board of directors at the general assembly meeting (Baldacchino et al., 2020). Supporting this argument, Shen et al. (2015) claim that many shareholders are interested only in increasing profits and maximizing their wealth.

2.2.5. Environmental disclosure

There are quite different requirements for disclosure in developing and developed markets. Company information is conveyed to relevant parties in a variety of ways: letters to shareholders, websites, newspapers, and published interviews. One main mechanism is companies' annual reports (Luo et al., 2018). The vast majority of investors count on a firm's annual reports for their investment decision-making. They provide a brief overview of companies and are more credible and timely sources than media reports. In the Jordanian market, there are three kinds of disclosure processes: voluntary disclosures, mandatory disclosures, and disclosures of the corporates' social responsibility.

In 1976, AFM ("Amman Stock Exchange", n.d.) law addressed the requirements of disclosure for Jordanian firms in general terms, as at the time, there was no extant information on any specific disclosure requirements for annual reports for companies listed on the ASE. In addition, there were no requirements in terms of the period and presentation of published financial information in Jordan (Khateeb, 2022; Alawneh & Alawneh, 2022; Saaydah, 2022). There is also an absence of enforcement tools for companies' compliance with the disclosure requirements, such as they were. Therefore, the quality of the disclosure environment for Jordanian firms has been classified as unsatisfactory. In sum, this is due to what has been described as an inadequate set of standards for auditing and accounting, without guidelines or legal requirements for Jordanian disclosure practices.

Through the Temporary Securities Law No. 23 of 1997, Jordan set out the Directives of Disclosure and Accounting and Auditing Standards, in which certain requirements for annual reports were listed, investors' encouraging trust and company accountability and transparency through compliance with international standards. This law was replaced by the Securities Law of 2002, delegating the ASE, JSC, and SDC greater authority to urge companies to conform to disclosure requirements. This reform meant that all listed companies had to provide their annual reports to the JSC within a specific time after the fiscal year's end.

Overall, voluntary, mandatory, and corporate social responsibility (CSR) disclosures gradually improved in Jordan due to successive reforms since 1997. Among these was the Environmental Protection Law No. 52 of 2006, which it compulsory for a firm to assess its influence on the environment and provide this in the annual report. As expected, compliance with mandatory disclosure exceeds that of the other two kinds of disclosure. Haddad et al. (2017) argue that cultural and social factors heavily influence voluntary disclosure. In developing countries, there is a tendency for societies to be "more secretive, conservative and based on statutory control, with little professional judgment compared to their counterparts of developed countries' (Haddad et al., 2009, p. 289). This thus accounts for little voluntary disclosure in these environments. In Jordan, there is insufficient information on social and environmental disclosure required by laws, regulations, and instructions, which is why relatively few CSR disclosures are made. For example, the 2017 CG instructions for shareholding listed firms addressed the idea of CSR in general terms, merely requiring each company to develop its policy about social responsibility for the local community and the environment, without establishing any particular content or type of disclosure about CSR activities. For this reason, Jordanian companies' commitment to environmental protection remains weak.

In sum, despite consistent improvement in Jordan, not all companies comply with regulations about disclosure. Institutional regulatory bodies still fail to mount pressure on corporates to comply with the above-mentioned principles, regulations, laws, and standards. In addition, some loopholes exist in these laws and regulations and there are insufficient qualified auditors and accountants to develop an effective basis for compliance, especially given the concentration of firm ownership.

2.3. The assessment of the model of Jordanian corporate governance

Using a desk-based appraisal of relevant documents, the current state of Jordanian firms' compliance with CG practices is measured. Gaps are identified between the nation's laws and regulations for CG compared with international CG principles and implementation practices. This assessment offers a comparative analysis of the quality and effectiveness of the country's legislation for CG, including enforcement mechanisms.

In Jordan, Shanikat and Abbadi (2011) and Cigna and Sigheartau (2017), have conducted such an assessment, focusing on Jordanian firms' compliance with CG principles. Shanikat and Abbadi (2011) approach this using the relevant law and what happens in practice (see Table 1).

Cigna and Sigheartau (2017) have also identified Jordanian banks' CG strengths and weaknesses, investigating five areas: rights of stakeholders, rights of shareholders, board structure functioning, and institutions, internal control, and transparency and disclosure. A five-level rating with the highestlevel conflating strong and very strong into a single level, and the lowest as very weak (further details are presented in Table 2).

VIRTUS

Principles	Assessment base		Domorks
Principies	In law	In practice Remarks	
The rights of shareholders	Widely covered	Widely practised	Shareholders participate in most important decisions except main asset sales. Shareholders' AGM rights are also mentioned, but there are no standard proxy forms and no provisions for postal voting.
The equitable treatment of shareholders	Partially covered	Partially practised	The controller sometimes acts on shareholders' complaints, but there is no formal complaint-resolution mechanism. There are solid regulations prohibiting insider trading. Related-party transaction rules are not clear.
The role of stakeholders in CG	Covered	Practised	Stakeholder rights are respected. Stakeholders have a number of legal Protections, which are widely covered in the Companies Law. Companies typically adopt performance enhancement measures, such as employee savings funds. Employees sometimes share ownership in some companies' issues.
The disclosure and transparency	Covered	Practised	Annual and semi-annual reports are provided, but only the annual report is required to be audited externally. Monitoring is limited only to quantity rather than quality of disclosures. There are no comprehensive and mandatory rules for corporate governance disclosure. Jordan has fully adopted the IFRS and ISA standards for accounting and audits.
The responsibility of the board	Covered partially	Practised	The board is liable for ensuring compliance with the law. In practice, there is no difference between the management and the board; generally, the chairman and CEO are the same person. Stakeholders' duties are not clear. The law and regulations determine specific standards related to functions that the board should fulfil. By the law, directors have a right to access all relevant information.

Table 1. The assessment of the Jord	lanian corporate governance model
-------------------------------------	-----------------------------------

Note: IFRS — International Financial Reporting Standards, ISA — International Standards on Auditing. Source: Shanikat and Abbadi (2011, p. 100).

Kamar and Selim (2020) issued a report under the auspices of the European Bank for Reconstruction and Development, evaluating that, overall, firms' board structure and functioning are weak, though designed as internal control systems. This report indicated that since 2007, reporting standards and financial auditing have weakened, with few improvements in firms' ethics in Jordan and that Jordan's regulations and governance are seen as impediments to attracting investment. It is claimed that policy instability and inefficient government bureaucracy are key problematic factors for a business

operating in Jordan. For example, the short lifespan of Jordanian parliaments has hindered the sustainability of effective government initiatives. Supporting these claims, incidents of uncertainties faced by the private sector are symptomatic of weak CG processes relevant to the way strategies, regulations, and policies are designed, coordinated, conducted, and updated. The conclusion is that foreign and domestic investors need a level of certainty, transparency, and accountability about government policy that might influence their decisions to commence and operate businesses.

Table 2. An overall assessment of the Jordanian corporate governance model

Key groups	Rating
The structure of the board and functioning, independent directors, board effectiveness, internal control, functioning and independence of the audit committee, control over related party transactions and conflict of interest, general shareholders' meeting, minority shareholders' protection and access to information, registration of shareholdings, and CG code	Weak
Board composition, non-financial information disclosure, and quality of internal and external audit	Fair/Weak
Board responsibilities, transparency and disclosure, disclosure of the external audit, reporting to the market and shareholders, quality of the internal control framework, and rights of shareholders	Fair
Protection against insider trading and self-dealing	Moderately strong
Financial information disclosure	Strong
Gender diversity at the board	Very weak

Source: Al-Msiedeen (2019).

2.4. The Jordanian institutional forces

In this study, Jordan's official institutional capacity for strengthening CG best practices by firms is evaluated. Key institutional forces that can exert pressure on firms to this end are explained in the following paragraphs, after a summarized introduction.

2.4.1. Jordan's economy — A brief sketch

Jordan's economy is among the smallest in the Middle East. It exports potash and phosphate but has insufficient water, oil, and other natural resources. Therefore, Jordan's government relies on external aid. Jordan faces a budget deficit and high unemployment, poverty, and government debt. After the COVID-19 shock, Jordan has started its recovery. Gross domestic product (GDP) grew from 1.6% in 2020 to 2.2% in 2021. In Q1 2022, growth rebounded to 2.5 %, supported by the reopening of the economy and the recovery of some sectors, notably tourism. However, higher commodity prices around the world led to an acceleration in headline inflation and labor market conditions remain challenging. For instance, unemployment in Q2 2022 is 22.6%, which is still above pre-pandemic levels (with women at 29.4% and youth under 25 years old at 46.1%). In addition, labor force participation remains low (33.5%) in Q2 2022, especially for women (14.2%), one of the lowest rates in the world. It is worth noting that the unfavorable global



context poses important risks, despite robust growth in exports and a solid rebound in travel receipts. At the end of 2021, total debt reached 113.7% in Jordan. Further, Jordan faces climate-related hazards, including precipitation decreases, temperature increases, and more incidents of drought. This weak growth can be traced to multiple external shocks, including regional conflicts and the influx of almost 1.3 million Syrian refugees (representing nearly 13% of the total Jordanian population) (World Bank, 2023).

In fact, economic transformation in Jordan remains contingent on identifying opportunities to conduct the reforms needed to encourage private sector-led growth and job creation. Among these is the government's program of economic priorities 2021–2023, which prioritizes key business environment reforms to support public-private and finance development partnerships of the investment environment and job creation. To support these reforms, Jordan initiated a new economic modernization in 2022, vision for targeting growth and opportunities in the following ten years, together with a plan for public sector modernization (World Bank, 2023).

Some public firm shares were earlier traded irregularly, through private brokers who bought, sold and tracked trading in the absence of an organized market for securities trading in Jordan. This system was faulty, inefficient, and laborious and the need for a well-organized market became recognized as crucial. The capital market plays a central role in the economic growth of any country.

According to Al-Khouri and Al-Ghazawi (2008), the volume of trading was very limited. In 1930 and 1931, Jordanian Tobacco and Cigarettes and the Arab Bank were established, respectively. In 1951, Jordanian Cement Factories were established in the south. In the early 1960s, bonds for some Jordanian firms were issued and traded (Al-Khouri & Al-Ghazawi, 2008).

As the number of public firms rose, facilitated trading, and the protection of investors' interests became more important, prompting the increased demand for a developed Jordanian securities market. This led to the government's institution of the AFM in 1976, Jordan's first securities market, in cooperation with the IFC, which began operations in 1978 as the only stock exchange in Jordan.

То develop securities market its to international standards, the Jordanian government decided that further reforms of the Jordanian market were needed. In 1997, the Temporary Securities Law was issued, a watershed moment for Jordan's capital market. This law's purpose was to restructure and regulate this capital market by separating legislative and supervisory roles from the executive role of the Jordanian securities market. In this way, the AFM was replaced by the ASE, JSC, and SDC. As mentioned earlier, the JSC was allocated legislative and supervisory functions, while the ASE and the SDC took over executive functions.

2.4.2. Amman Stock Exchange

The ASE, established by the Temporary Securities Law No. 23 of 1997, is considered a non-profit organization with administrative and financial independence. The main reason behind setting up the ASE was likely Jordan's privatization program. On March 11, 1999, the ASE embarked on its operations with 151 listed companies and market capitalization reached nearly USD5,844.2 million by the end of 1999. In 2017, it was registered as a public shareholding firm, completely owned by the Jordanian government with the name "The Amman Stock Exchange Company" (ASE Company). The ASE Company is the factual and legal successor to the ASE and is managed by a board of seven directors appointed by the Council of Ministers, all of whom are non-executive and non-independent. This board established four committees, namely, the Nomination and Remuneration Committee, Governance Committee, Auditing Committee, and Risk Management Committee. The Governance Committee is responsible for preparing and submitting the governance report to the board to ensure ASE compliance with the CG code issued by the Companies Control Department (CCD) and the company's disclosure instructions issued by the JSC.

As indicated above, it may be argued that a conflict exists between the international CG code and the CG code issued by the CCD, with the ASE Company conforming with the latter. In 2021, the ASE performed a set of producers related to environmental, social, and governance. As of 2020, all companies listed on the ASE must issue a sustainability report regarding the CG, and their social and environmental responsibilities. Interestingly, the ASE launched an initiative related to the disclosure of information related to climate change, in cooperation with the IFC.

2.4.3. Jordan Securities Commission

The JSC is an official independent government body reporting directly to the prime minister, with supervisory and legislative functions (Alhusban et al., 2020). Its purpose is to administer and develop Jordan's capital market to ensure transparency and efficiency and protect it from any likely threats. The JSC is subject to the supervision of external and internal audits and the Jordan's Audit Bureau. In addition, it complies with the standards of the International Organization of Securities Commissions (IOSCO). The Securities Law of 2002 further strengthened Temporary Securities Law No. 23 of 1997, giving the JSC the powers to control and organize the issuance of securities, registration, matters concerning licensing and disclosure, and instructions to improve CG practice. In 2017, the JSC revised the JCGC of 2009 by requiring all firms listed on the ASE to include a governance report in their annual reports.

Unfortunately, much less attention has been given to the disclosure of information related to climate change in these JCGC 2017 revisions, although the current challenges facing Jordan are climate-related hazards. There is no mandatory formal legislation or regulations for Jordanian companies to disclose information related to climate change in their annual reports.

VIRTUS 197

2.4.4. Securities Depository Centre

In 1999, the SDC was instituted by the Temporary Securities Law No. 23 of 1997 as a non-profit entity with administrative, financial, and legal autonomy. The SDC's purposes are to protect the securities ownership and process registration ownership of securities among brokers. It seeks to reinforce the confidence of investors, assisting them in obtaining information concerning their investments easily and quickly. Therefore, it plays a vital role in limiting the dangers in the settlements of trading transactions. According to Section 77 of the Securities Law of 2002, the SDC in Jordan is the only entity in the market to: register, safely keep, and transfer securities ownership; deposit the ownership of securities; clear and settle the ownership of securities. The SDC registers and deposits stocks issued by public shareholding companies and registers treasury bonds and bills, and individual savings bonds issued by the Jordanian government. Thus, the membership in SDC is mandatory for public shareholding firms (i.e., the insurance, services, industrial, and banking sectors), custodians, brokers, and any other entities as determined by the JSC's Board of Commissioners.

2.4.5. Insurance Commission

The Insurance Regulatory Act No. 33 of 1999 set up the IC as a government entity with administrative and financial independence. The IC aims to regulate, monitor, and enhance the insurance sector and, in doing so, contribute to protecting the Jordanian economy. It moved slowly in communicating its instructions. In 2003, the IC provided Instructions on Accounting Policies to be Considered by Insurance Firms, together with a group of forms required to prepare financial reports for a standardized style for the production of such information (Al-Tal, 2014). The IC issued Corporate Governance Instructions in 2006, in accordance with the provisions of the Insurance Regulatory Act No. 33 of 1999, Section 45(B), applying specifically to the insurance companies' sector, specifying the responsibilities, duties, executive management, internal auditors, the audit committee, and roles of directors. The IC is responsible for following up on insurance companies' annual reports, which must be audited by the company's external auditor and displayed within specified times.

2.4.6. Companies Control Department

The Companies Law No 22 of 1997 set the framework for the establishment of the CCD in 2003 as an independent organization. The CCD's main purpose is to register various types of firms in the Jordanian market and attempt to influence them by improving the procedures for such registration, thereby reinforcing the internal and external mechanisms for control over firms in a general way. It does this by monitoring public shareholding companies' disclosures. The Companies Law No 22 of 1997 grants the CCD authority to monitor the remuneration and ownership of the board of directors by checking public shareholding companies' reports. In 2012, the CCD issued the code of the CG, which is specifically applicable to limited liability firms, public shareholding corporates not listed on the ASE, the private corporate sector, private shareholding firms, and limited liability firms that are not for profit. As indicated above, this code contains five main sections related to the board of directors (management committee roles and responsibilities), the rights of shareholders and stakeholders, disclosure, transparency, and the control environment. However, as all members of the board are required to be shareholders in the same company by the Companies Law No 22 of 1997 this appears to be substantially inconsistent and in conflict with CG's best practice in terms of the independence of board directors.

2.4.7. Central Bank of Jordan

The CBJ was established by virtue of the CBJ Law of 1959 and started its operations in 1964. The CBJ is a public entity with legal, administrative, and financial autonomy. It seeks to maintain monetary and financial stability, thereby eventually contributing to the creation of economic growth in Jordan. The CBJ enacted the JCGC for Banks in 2007, which was based on the principles of the OECD and the guidelines enacted by the Basel Committee on Banking Supervision, to enhance CG best practices. For this, the CBJ reiterated four key principles found in those other documents (see the JCGC for banks of 2007). This ICGC for banks was revised in 2016 to include the responsibilities of the board, board composition, stakeholders' interests, conflicts of interest, and disclosure and transparency.

2.5. The legal environment in Jordan

The legal framework relating to the Jordanian company sector consists of specific laws and numerous legislative mechanisms, such as orders, regulations, rules, instructions, notifications, and circulars, issued by formal bodies like the JSC, ASE, CCD, and other related governmental and private agencies, including the Jordan Chamber of Industry. The legal environment plays a key role in formulating and enforcing the best practices of CG principles within an organization's structure and affecting its behavior (DiMaggio & Powell, 1991).

Overall, in most developing countries, including Jordan, the institutional setting and the legal environment are qualitatively weak (La Porta et al., 2000). More recently, Haddad et al. (2017) note scant legal requirements for disclosure in Jordan. Hence, with such a weak legal base, there is limited scope for enforcement of such disclosure in the Jordanian market. Such weak legal enforcement leads to low compliance by companies, and indeed noncompliance with the Securities Law.

2.6. The professions of accounting and auditing in Jordan

The accounting profession is generally thought to be among the most significant ones affecting the national economy, with its roots in the Jordanian legal system, a hybrid of civil law and Islamic legal principles (Haddad et al., 2017). Historically, commerce was managed by the Ottoman Commercial Code enacted between 1849 and 1850 in what was then known as Transjordan. This was replaced after Jordan's independence in 1946. In 1964, Jordan issued its first company law, which was used in both the East and West Banks of Jordan. The accounting profession in Jordan has been strictly monitored by the Accounting Professional Council (Al-Akra et al., 2009). In 1985, an Audit Bureau was set up, a defining moment in managing accounting practice in Jordan (Solas, 1994). Through the Act of 1987, the Jordanian Association of Certified Public Accountants (JACPA) was founded (Atmeh, 2016). In 1992, it joined the International Federation of Accountants. Until 1997, no independent formal or legal entity was involved with developing accounting and auditing standards in the country (Nassar et al., 2013). Instead, the Ministry of Industry and Trade supervised these professions, with JACPA having a limited role (Al-Akra et al., 2010). Due to the Companies Law and the Securities Law, in 1997 and 2002, respectively, all shareholding firms in Jordan needed to adopt the full versions of the International Accounting Standards (IAS) and IFRS (Al-Akra et al., 2009). In 2003, a new law was passed to develop the professions of accounting and auditing and guarantee the successful application of IAS and IFRS in Jordanian companies. In 2004, the High Council for Accounting and Auditing was established by the government. It promoted JACPA, by this stage attached to the Public Auditing Profession Board. In this way, JACPA obtained more authority, including the responsibility to formulate by-laws (Haddad et al., 2017). However, much less attention has been given to auditor independence and any relevant code of ethics. This proved to have severe ramifications in that there is an obvious absence of enforcement tools.

While the auditing profession had purportedly been managed by Law No. 10 of 1961 (Haddad et al., 2017), enforcement was lax. In 1985, Auditing Profession Practice Law No. 32 came into being, one important step in developing the profession and its later support of JACPA, established as mentioned above, through JACPA's Act of 1987. The Auditing Profession Practice Law No. 32 also assisted in later setting up the High Council for Accounting and Auditing by making membership in JACPA obligatory for auditors; thus, the Act was designed to organize the auditing profession, to some extent, by identifying who worked as auditors in Jordan.

3. RESEARCH METHODOLOGY

The main objective of this research is to provide an overview of the current practices of CG and highlight the roles of main institutional forces in strengthening the existing CG practices in emerging markets by considering Jordan as a case study based on the literature in this area. Following the previous studies, such as Petticrew and Roberts (2006), this study can be described as a review paper. To do this end, archival data is used for this study (e.g., academic literature, reports published by the main institutions, financial reports published by listed companies and revisions of relevant legislations and laws). Accordingly, the main issue in this research will be studied along with a discussion on the main institutional forces that reinforce the existing CG practices in Jordan. Therefore, this will include Jordanian codes of CG, company law, the web pages of the main institutions and legal legislation environment in order to bring accountability to the Jordanian corporate sector.

4. THEORETICAL FRAMEWORK

In order to explain CG and its issues, researchers have extensively depended on resource dependence theory, stewardship theory, agency theory, transaction cost theory, and institutional theory (Steinfeld, 2023; Marashdeh et al., 2021; Löhde et al., 2021; Jhunjhunwala, 2023). To understand CG and its institutional changes in Jordan, institutional theory is adopted as an explanatory device. Institutional theory has been described by North (2005) as a set of formal and informal rules, standards, values, norms, and behaviors that influence business activity. As noted by del Carmen Briano-Turrent and Rodríguez-Ariza (2016), CG practice may be notably affected by institutional forces (isomorphic pressures) (e.g., the institutional, regulatory, and legal environment, cultural aspects, financial markets, and the economic environment) which constitute the institutional framework (Scott, 2005). Hence, Jordanian companies have to take these pressures into account. This theory is reflected through many approaches, and researchers focus on different factors and changes in an organization (DiMaggio & Powell, 1991). Among these factors are regulatory, technological and political complexities (Greenwood & Hinings 1996). Thus, companies have to embrace these changes to survive.

Institutional theory can be classified as either "old" or "new" (Aksom et al., 2020; Burch & Crowson, 2020; DiMaggio & Powell, 1991; Greenwood & Hinings, 1996). Institutionalism's original focus was on coalitions, competing values, power, influence, and informational structures, whereas the new version concentrates on the legitimacy process, with the assumption that the key institutional goal is survival (Fogarty, 1996; Oliver, 1997). This research is underpinned by the new version. also termed new institutional sociology (NIS) (Meyer & Rowan, 1977; DiMaggio & Powell, 1983; Scott, 2005). In the words of Meyer and Rowan (1977), "institutional rules function as myths which organizations incorporate, gaining legitimacy, resources, stability, and enhanced survival prospects" (p. 340).

As argued by Fogarty (1996), organizational survival depends on its congruence with the prevailing principles or values for appropriate behavior. Organizations adopt such norms and behavior to legitimize their existence, ending to institutional pressures and the market; influenced by legitimization, corporations will adopt a similar structure, a process known as "institutional isomorphism" (Meyer & Rowan, 1977; DiMaggio & Powell, 1991). DiMaggio and Powell (1983) contended that such institutional isomorphic modification takes place in three ways (coercively, mimetically, and normatively). Coercive isomorphism stems from political influence in which formal and/or informal pressures are placed on some organizations through other organizations. DiMaggio and Powell (1983) indicated that in some cases, "organizational change is a direct response to government mandate: manufacturers adopt new pollution control technologies to conform to environmental regulations; non-profits maintain accounts, and hire accountants, in order to meet tax



law requirements; and organizations employ affirmative-action officers to fend off allegations of discrimination" (p. 150). Fogarty (1996) pointed out that the state, through delegation or its own action, becomes dominant in the coercion of companies through its control over resources or its definitions of the public interest.

Mimetic isomorphism stems from uncertainty in organizations (DiMaggio & Powell, 1983). Selznick (1996) claimed that organizations are sensitive to the changes occurring around them. They thus start to imitate such changes in other organizations in the same field that are perceived as more legitimate and successful (DiMaggio & Powell, 1983). Furthermore, uncertainty issues in organizations are a powerful pushing force that makes them attempt to respond to environmental changes to solve a problem. It may be argued that the process of imitation may encourage organizational sustainability and legitimacy.

Finally, normative isomorphism results from the pressure of professionalism. As DiMaggio and Powell (1983) claimed, "professional power is as much assigned by the state as it is created by activities of the professions" the (p. 153). Professionals are believed to hold particular moral values which they disseminate across organizations by adhering to a relevant code of ethics (Fogarty & Rogers, 2005). This supports the prevailing belief that professional bodies require their members to uphold certain values in their work (Abdullah et al., 2018). Therefore, achieving institutional isomorphism in these ways supports legitimacy on the one hand and, on the other, empowers organizations to continue their activities.

Researchers have extensively relied on NIS perspectives in explaining CG and its issues. NIS is considered relevant for analyzing organizations faced with uncertainties. Accordingly, they compete for institutional and political legitimacy and market position (Tsamenyi et al., 2006). Interestingly, international donor agencies have shown increased interest in CG in Jordan (e.g., the OECD, IFC and World Bank). Consequently, various CG reforms have become increasingly significant in Jordan's struggle for reinforced and sustained economic development. Part of these reforms involve the capacity building of the institutional environment and strengthening legitimacy.

5. RESULTS AND DISCUSSION

This research offers some insight into current CG practices in Jordan, a far less advanced institutional context than those in which the ISA and IFRS were initially devised. This evaluation of Jordan's institutional capability to reinforce CG practices points to important shortcomings in the relevant institutions and the major challenges and issues concerning CG at the company level. The theoretical foundation for this study is new institutional sociology. Jordan is struggling toward comprehensive institutional development to strengthen good CG practices in line with international CG best practices for listed companies.

Since the early 1997s, many reforms have been undertaken. The initial work of the AFM was developed by its replacement with three institutions (the ASE, JSC, and SDC) and the Companies Law No 22 of 1997 and the Securities Law of 2002 were enacted. This meant a revision of Jordan's capital market regulations and laws, separation of the legislative and supervisory roles from the executive role in the Jordanian securities market, adoption of ISA and IFRS, and a revision of the JCGC for listed firms. Overall, this developing institutional environment promotes CG in Jordan. However, further reforms would benefit the efforts made to date.

In general, the current institutional regulatory regime fails to pressure companies to abide by the laws, regulations, principles, and standards outlined above, which leads to substantial noncompliance. As a result, corporations in Jordan, in many cases, do not follow the guidelines associated with CG best practices. It can be said that the executives and directors in Jordanian boardrooms who belong to one family have a tendency to behave opportunistically, for their own interests, at the expense of those of their companies. It is argued that the predominance of family-owned Jordanian firms has engendered serious challenges to effective governance in Jordan at the company level. One plausible explanation is that families hinder governmental institutions' actions in their attempts to enforce the relevant laws and regulations. Families' political relationships typically directly oppose the state's legal authority (Uddin & Choudhury, 2008). Yet another explanation by Jalilian et al. (2007) is that building an effective and efficient regulatory environment in developing countries does not merely involve technical aspects regarding the most appropriate regulatory tools; it relevant to is the quality of regulatory institutions and their capability to strengthen CG best practice.

Arguably, the various regulatory bodies in Jordan also suffer from a lack of professional ethics and a shortage of talented and skilled managers. This is an institutional problem. Therefore, it can be argued that although Jordan has apparently adopted CG best practices consistent with the international CG best practice for institutional legitimacy, in fact, there are indications of weak CG processes concerning the design of relevant strategies, regulations and policies, and how they are coordinated, operationalized and updated. A number of initiatives to simplify the regulatory framework is underway or planned; however, progress has been very limited so far. instability and inefficient government Policy bureaucracy are key problematic factors for a business operating in Jordan. For the past several vears, there has been a decrease in market capitalization and the number of listed firms in Jordan. In the last five years in Jordan, it has become less easy for firms to obtain information about changes in government policies, legislation, and regulations influencing their activities. Consequently, regulatory quality in Jordan has slightly worsened over the past decade.

In brief, additional key reforms are needed for some aspects of CG in Jordan. Rigorously devised regulation is needed to deal with opportunistic people who benefit in a context where the market plays a limited role. Further, educating investors may assist in ensuring that managers work in the company's best interests, ultimately making the company accountable. The features of CG practices within a given country are heavily affected by the "forces aimed at increasing their efficiency,



and legitimization effects due to path dependence" (Zattoni & Cuomo, 2008, p. 1). Therefore, the reforms must be intelligently designed to boost the governance environment in Jordan, keep up with its changes and complexity, and finally, promote an environment of accountability among Jordanian firms.

6. CONCLUSION

The current study aims to provide a comprehensive view of the current practice of CG in the emerging market, with a specific focus on Jordan. It also shows a comprehensive insight into whether the main institutional forces are able to reinforce the current practices of CG in the Jordanian corporate sector. The main results of the study indicate that the institutional settings are all very weak. This leads to the Jordanian corporate sector failing to follow the good practices of CG in many instances.

The results of this study have valuable implications for regulators, authorities, policymakers, practitioners, and managers of companies in Jordan. Firstly, policymakers could benefit from the study's results by selecting which legislative aspects need development and formulating clear regulations and rules concerning the CG, and eventually development environment of CG in Jordan on one hand, and, on the other, promote the roles formal institutional forces in strengthening the existing CG practices in Jordan. This leads to an increase in the level of compliance among Jordanian companies. Indeed, the importance of this study for relevant parties is that it has critical implications for the present reform movement of CG practices in Jordan. Secondly, the research's results call for official or

legal bodies to pay attention to encourage the roles of main institutions in Jordan. Finally, to improve the current structure of CG in Jordan, this study calls on the related parties to benefit from international standards of CG may assist in correcting and formulating the governance system in any country around the world by adopting some beneficial features of CG which exist in developed countries. Therefore, this research makes some recommendations depending on the best practices of CG around the world, in keeping with the current system of CG in Jordan. This study makes a critical contribution to the governance literature and alleviates the scarcity of studies on CG in developing markets.

Just like any other study, this study also suffers from some limitations, which opens the way for possibly fruitful future research opportunities. Firstly, although the findings of this study are specific to the Jordanian scenario, the similarity of the Jordanian governance model with those of other emerging markets points out that an extended investigation of the subject in the Middle Eastern countries is recommended. For example, future research can investigate CG across Arab countries with similarly shared institutional settings, cultures, economies and financial infrastructures. Secondly, this study can be described as a review paper. However, it is suggested that future research in this area may explore the other aspects of CG at the corporate level by adopting different methodological perspectives. Finally, the institutional factors influencing the good practice of CG in Jordan are relatively new, and therefore, there is a scarcity of studies on this topic in emerging economies like Jordan.

REFERENCES

- 1. Abdullah, Z., Chizema, A., & Chandren, S. (2018, October 16-17). *The perspective of corporate governance reformation from the lens of institutional theory* [Paper presentation]. 5th International Conference on Accounting Studies (ICAS 2018), Penang, Malaysia. https://repo.uum.edu.my/id/eprint/26077
- 2. Abdullatif, M., Alhadab, M., & Mansour, I. (2019). Determinants of related party transactions in Jordan: Financial and governance factors. *Australasian Accounting, Business and Finance Journal, 13*(1), 44–75. http://doi.org/10.14453/aabfj.v13i1.4
- 3. Act of the Jordanian Association of Certified Public Accountants issued pursuant to Article 18 of the Auditing Profession Law No. 32 of 1985. (1987). Jordanian Association of Certified Public Accountants (JACPA). https://jordanianlaw.com/%D8%A7%D9%84%D8%A3%D9%86%D8%B8%D9%85%D8%A9/%D9%86%D8%B8%D8%A7%D9%85%D8%AC%D9%85%D8%AC%D9%85%D8%AF%D9%85%D8%AF%D9%82%D9%82%D9%84.*%D8%A7%D9%84%D8%AAD%D8%B3%D8%A7%D8%A8%D8%AA-%D8%A7%D9%84%D9%82%D8%A7%D9%86%D9%88%D9%86%D9%8A%D9%86%D9%86%D9%86%D9%86%D9%8A%D9%86%D9%86%D9%86%D9%86%D9%8A%D9%86%D9%86%D9%86%D9%86%D9%8A%D9%86%00%9%96%0%9%96%9%9%9%9%9%9%9%
- 4. Aguilera, R. V., & Jackson, G. (2010). Comparative and international corporate governance. *Academy of Management Annals,* 4(1), 485–556. https://doi.org/10.5465/19416520.2010.495525
- 5. Aksom, H., Zhylinska, O., & Gaidai, T. (2020). Can institutional theory be refuted, replaced or modified? *International Journal of Organizational Analysis, 28*(1), 135–159. https://doi.org/10.1108/IJOA-02-2019-1666
- 6. Al-Akra, M., & Hutchinson, P. (2013). Family firm disclosure and accounting regulation reform in the Middle East: The case of Jordan. *Research in Accounting Regulation*, *25*(1), 101–107. https://doi.org/10.1016/j .racreg.2012.11.003
- 7. Al-Akra, M., Ali, M. J., & Marashdeh, O. (2009). Development of accounting regulation in Jordan. *The International Journal of Accounting*, *44*(2), 163–186. https://doi.org/10.1016/j.intacc.2009.03.003
- 8. Al-Akra, M., Eddie, I. A., & Ali, M. J. (2010). The influence of the introduction of accounting disclosure regulation on mandatory disclosure compliance: Evidence from Jordan. *The British Accounting Review*, *42*(3), 170–186. https://doi.org/10.1016/j.bar.2010.04.001
- Alawneh, A., & Alawneh, A. (2022). The effect of implementing the International Financial Reporting Standards (IFRS) on Jordanian financial market reporting effectiveness. *International Journal of Information, Business and Management, 14*(3), 93–102. https://www.proquest.com/openview/d21301c34becf5bfdd63b9dcaf07a360/1?pqorigsite=gscholar&cbl=2032142
- 10. Al-Begali, S. A. A., & Phua, L. K. (2023). Accruals, real earnings management, and CEO demographic attributes in emerging markets: Does concentration of family ownership count? *Cogent Business & Management, 10*(2), Article 2239979. https://doi.org/10.1080/23311975.2023.2239979

VIRTUS

- 11. Alhusban, A. A. A., Haloush, H. A., Alshurafat, H., Al-Msiedeen, J. M., Massadeh, A. A. M., & Alhmoud, R. J. (2020). The regulatory structure and governance of forensic accountancy in the emerging market: Challenges and opportunities. *Journal of Governance & Regulation*, *9*(4), 149–161. https://doi.org/10.22495/jgrv9i4art13
- 12. Al-Khouri, R., & Al-Ghazawi, N. (2008). The effect of electronic trading on market volatility and liquidity in emerging markets: Evidence from Amman Stock Exchange. *Journal of Derivatives & Hedge Funds, 14*, 222–236. https://doi.org/10.1057/jdhf.2008.21
- 13. Alkurdi, A., & Mardini, G. H. (2020). The impact of ownership structure and the board of directors' composition on tax avoidance strategies: Empirical evidence from Jordan. *Journal of Financial Reporting and Accounting*, *18*(4), 795–812. https://doi.org/10.1108/JFRA-01-2020-0001
- 14. Al-Msiedeen, J. M. (2019). *The influence of corporate governance practices on firm performance and efficiency: Evidence from Jordan* [Doctoral dissertation, University of Southern Queensland]. https://doi.org/10 .26192/xe8q-j527
- Al-Msiedeen, J. M., & Al Sawalqa, F. A. (2021). Ownership structure and CEO compensation: Evidence from Jordan. Asian Economic and Financial Review, 11(5), 365–383. https://doi.org/10.18488/journal.aefr .2021.115.365.383
- Al-Msiedeen, J. M., Rashid, A., & Syed, S. (2018). Board independence and firm performance: Evidence from Jordan. In *Proceedings of Sydney International Business Research Conference 2018* (pp. 105–121). Australian Academy of Business Leadership. http://www.aabl.com.au/aablConference/public/documents/pdf/2019_09 _30_10_41_38_AABL_P129_Full_Paper.pdf
- 17. Al-Najjar, B., & Clark, E. (2017). Corporate governance and cash holdings in MENA: Evidence from internal and external governance practices. *Research in International Business and Finance, 39*(Part A), 1–12. https://doi.org/10.1016/j.ribaf.2016.07.030
- Alshirah, M. H., Alshira'h, A. F., & Lutfi, A. (2022). Political connection, family ownership and corporate risk disclosure: Empirical evidence from Jordan. *Meditari Accountancy Research*, 30(5), 1241–1264. https://doi.org/10.1108/MEDAR-04-2020-0868
- 19. Al-Tal, Y. A. (2014). *The role of the Amman Stock Exchange in Jordanian corporate governance* [Doctoral dissertation, University of Western Sydney]. http://handle.uws.edu.au:8081/1959.7/uws:29890
- 20. Altawalbeh, M. A. F. (2020). Corporate governance mechanisms and firm's performance: Evidence from Jordan. *Accounting and Finance Research*, *9*(2), 11–22. https://doi.org/10.5430/afr.v9n2p11
- 21. Amman Stock Exchange: General information. (n.d.). The Official Site of the Jordanian e-Government. https://portal.jordan.gov.jo/wps/portal/Home/GovernmentEntities/Agencies/Agency/Amman+Stock+Exchange ?current=true&nameEntity=Amman%20Stock%20Exchange&entityType=otherEntity&lang=en&isFromLangChange=yes
- 22. Atmeh, M. (2016). The players in the accountancy profession and their roles: The case of Jordan. *International Journal of Business and Management*, *11*(3), 64–72. https://doi.org/10.5539/ijbm.v11n3p64
- 23. Auditing Profession Practice Law No. 10 of 1961. (1961). https://jordanianlaw.com/%D8%A7%D9%84%D9 %82%D9%88%D8%A7%D9%86%D9%8A%D9%86/%D9%82%D8%A7%D9%86%D9%88%D9%86-%D9%85%D8%A2 %D9%88%D9%84%D8%A9-%D9%85%D9%87%D9%86%D8%A9-%D8%AA%D8%AF%D9%82%D9%8A%D9%82-%D8%A7%D9 %84%D8%AD%D8%B3%D8%A7%D8%A8%D8%A7%D8%AA/
- 24. Auditing Profession Practice Law No. 32 of 1985. (1985). http://site.eastlaws.com/GeneralSearch/Home /ArticlesTDetails?MasterID=1422653
- 25. Baldacchino, P. J., Duca, N., Tabone, N., & Grima, S. (2020). Corporate governance transparency in small listed entities: The case of Malta. *European Research Studies Journal, 23*(2), 23-44. https://doi.org/10 .35808/ersj/1578
- 26. Burch, P., & Crowson, R. L. (2020). Local schooling and organizational change: New insights from the perspective of institutional theory. *Peabody Journal of Education*, *95*(4), 331–335. https://doi.org/10.1080/0161956X.2020.1800171
- 27. Central Bank of Jordan (CBJ). (1971). Central Bank of Jordan Law No. 23 of 1971 and amendments thereto. *Official Gazette, 2301.* https://www.cbj.gov.jo/EchoBusV3.0/SystemAssets/a6d37e65-4ed7-41d0-8799-609 62f082f77.pdf
- 28. Cigna, G. P., & Sigheartau, A. (2017). *Corporate governance in transition economies: Jordan country report*. European Bank for Reconstruction and Development. https://www.ebrd.com/sites/Satellite?c=Content&cid =1395251705258&pagename=EBRD%2FContent%2FDownloadDocument
- 29. Companies Law No. 22 of 1997. (1997). *Official Gazette, 4204*. https://www.jsc.gov.jo/Uploads/Files/%D9%82 %D8%A7%D9%86%D9%88%D9%86%20%D8%A7%D9%84%D8%B4%D8%B1%D9%83%D8%A7%D8%AA%20%D8%B3%D9 %83%D8%A7%D9%86%D8%B1.pdf
- 30. Cornett, M. M., Marcus, A. J., Saunders, A., & Tehranian, H. (2007). The impact of institutional ownership on corporate operating performance. *Journal of Banking & Finance*, *31*(6), 1771–1794. https://doi.org/10.1016 /j.jbankfin.2006.08.006
- 31. Davis, G. F. (2005). New directions in corporate governance. *Annual Review of Sociology*, *31*, 143-162. https://doi.org/10.1146/annurev.soc.31.041304.122249
- 32. del Carmen Briano-Turrent, G., & Rodríguez-Ariza, L. (2016). Corporate governance ratings on listed companies: An institutional perspective in Latin America. *European Journal of Management and Business Economics*, *25*(2), 63–75. https://doi.org/10.1016/j.redeen.2016.01.001
- 33. DiMaggio, P. J., & Powell, W. W. (1983). The iron cage revisited: Institutional isomorphism and collective rationality in organizational fields. *American Sociological Review, 48*(2), 147–160. https://doi.org/10.2307/2095101
- 34. DiMaggio, P. J., & Powell, W. W. (1991). Introduction. In W. W. Powell & P. J. DiMaggio (Eds.), *The new institutionalism in organizational analysis*. University of Chicago Press. https://web.stanford.edu/~woodyp /papers/dimaggioandpowell_intro.pdf
- 35. Environmental Protection Law No. 52 of 2006. (2006). The Ministry of the Environment. https://jo.chmcbd.net/cooperation-1/laws-and-regulations/environmental-protection-law-no.-52-of-2006
- 36. Epps, R. W., & Cereola, S. J. (2008). Do institutional shareholder services (ISS) corporate governance ratings reflect a company's operating performance? *Critical Perspectives on Accounting*, *19*(8), 1135–1148. https://doi.org/10.1016/j.cpa.2007.06.007

VIRTUS 202

- 37. Fogarty, T. J. (1996). The imagery and reality of peer review in the US: Insights from institutional theory. *Accounting, Organizations and Society, 21*(2–3), 243–267. https://doi.org/10.1016/0361-3682(95)00001-1 38. Fogarty, T. J., & Rogers, R. K. (2005). Financial analysts' reports: An extended institutional theory evaluation.
- Accounting, Organizations and Society, 30(4), 331-356. https://doi.org/10.1016/j.aos.2004.06.003
- 39. Greenwood, R., & Hinings, C. R. (1996). Understanding radical organizational change: Bringing together the old and the new institutionalism. The Academy of Management Review, 21(4), 1022-1054. https://doi.org/10 .2307/259163
- 40. Haddad, A. E., AlShattarat, W. K., & Nobanee, H. (2009). Voluntary disclosure and stock market liquidity: Evidence from the Jordanian capital market. International Journal of Accounting, Auditing and Performance Evaluation, 5(3), 285-309. https://doi.org/10.1504/IJAAPE.2009.026629
- 41. Haddad, A. E., Sbeiti, W. M., & Qasim, A. (2017). Accounting legislation, corporate governance codes and disclosure in Jordan: A review. International Journal of Law and Management, 59(1), 147-176. https://doi.org/10.1108/IJLMA-07-2016-0064
- 42. Insurance Regulatory Act No. 33 of 1999. (1999). Jordan Insurance Federation. https://www.sdc.com.jo /arabic/images/stories/pdf/Insurance.pdf
- 43. Insurance Regulatory Law No. 33 of 1999. (1999). https://jordanlegalcenter.blogspot.com/2016/06/33-1999.html
- 44. Iswaissi, H., & Falahati, K. (2017). Challenges to corporate governance practices: Case study of Libyan commercial banks. Corporate Governance and Sustainability Review, 1(1), 33-41. https://doi.org/10 22495/cgsrv1i1p3
- 45. Jalilian, H., Kirkpatrick, C., & Parker, D. (2007). The impact of regulation on economic growth in developing countries: A cross-country analysis. World Development, 35(1), 87-103. https://doi.org/10.1016/j worlddev.2006.09.005
- 46. Jhunjhunwala, S. (2023). Theoretical perspectives of corporate governance: The foundation. In S. Jhunjhunwala (Ed.), Corporate governance: Creating value for stakeholders (pp. 53-72). Springer. https://doi.org/10.1007/978-981-99-2707-4 3
- 47. Jordan Securities Commission (JSC). (2017). Instructions of corporate governance for shareholding listed companies for the year 2017. https://www.jsc.gov.jo/Uploads/Files/corporate%20governance%20instructions %202017%20new.pdf
- 48. Kamar, B., & Selim, R. (2020). Jordan diagnostic 2020. European Bank for Reconstruction and Development. https://www.ebrd.com/country-diagnostic-paper-jordan.pdf
- 49. Kanakriyah, R. (2021). The impact of board of directors' characteristics on firm performance: A case study in Jordan. The Journal of Asian Finance, Economics and Business, 8(3), 341-350. https://doi.org/10.13106 /iafeb.2021.vol8.no3.0341
- 50. Khateeb, A. (2022). The effect of electronic disclosure of financial statements twhe market value of the stock prices listed in Amman Stock Exchange (ASE). In Proceedings of the 87th International Scientific Conference on Economic and Social Development — "Economics, Management, Finance and Banking" (pp. 394-401). Varazdin Development and Entrepreneurship Agency (VADEA). https://www.proquest.com/openview/d28b27c047d 334911b8525a06e16e32a/1.pdf?pq-origsite=gscholar&cbl=2033472
- 51. La Porta, R., Lopez-de-Silanes, F., Shleifer, A., & Vishny, R. (2000). Investor protection and corporate governance. Journal of Financial Economics, 58(1-2), 3-27. https://doi.org/10.1016/S0304-405X(00)00065-9
- 52. Löhde, A. S. K., Campopiano, G., & Calabrò, A. (2021). Beyond agency and stewardship theory: Shareholdermanager relationships and governance structures in family firms. Management Decision, 59(2), 390-405. https://doi.org/10.1108/MD-03-2018-0316
- 53. Luo, C., Li, M., Peng, P., & Fan, S. (2018). How does internet finance influence the interest rate? Evidence from Chinese financial markets. Dutch Journal of Finance and Management, 2(1), Article 1. https://doi.org/10 .20897/difm/89590
- 54. Maassen, G. F. (1999). An international comparison of corporate governance models. Spencer Stuart.
- 55. Mansour, M., Hashim, H. A., Almaqtari, F. A., & Al-ahdal, W. M. (2023). A review of the influence of capital structure on the relationship between corporate governance and firm performance. International Journal of Procurement Management, 17(1), 79-105. https://doi.org/10.1504/IJPM.2023.130269
- 56. Marashdeh, Z., Saidat, Z., Alkhodary, D., & Al-Haddad, L. (2021). Agency theory and the Jordanian corporate environment: Why a single theory is not enough. Academy of Accounting and Financial Studies Journal, 25(5). https://www.researchgate.net/publication/352212484_AGENCY_THEORY_AND_THE_JORDANIAN_CORPORATE ENVIRONMENT_WHY_A_SINGLE_THEORY_IS_NOT_ENOUGH
- 57. Meyer, J. W., & Rowan, B. (1977). Institutionalized organizations: Formal structure as myth and ceremony. American Journal of Sociology, 83(2), 340-363, https://doi.org/10.1086/226550
- 58. Nassar, M., Al-Khadash, H., & Mah'd, O. (2013). Accounting education and accountancy profession in Jordan: The current status and the processes of improvement. Research Journal of Finance and Accounting, 4(11), 107-119. https://www.researchgate.net/publication/259786020_Accounting_Education_and_Accountancy_Profession_in _Jordan_The_Current_Status_and_the_Processes_of_Improvement
- 59. North, D. C. (2005). Understanding the process of economic change. Princeton University Press. https://doi.org /10.1515/9781400829484
- 60. Oliver, C. (1997). Sustainable competitive advantage: Combining institutional and resource-based views. Strategic Management Journal, 18(9), 697-713. https://doi.org/10.1002/(SICI)1097-0266(199710)18:9<697::AID -SMI909>3.0.CO:2-C
- 61. Pearce, J. A., II. (2015). The rights of shareholders in authorizing corporate philanthropy. Villanova Law Review, 60(2), 251–282. https://digitalcommons.law.villanova.edu/vlr/vol60/iss2/1/
- Petticrew, M., & Roberts, H. (2006). Systematic reviews in the social sciences: A practical guide. Wiley. 62. https://doi.org/10.1002/9780470754887
- Rashid, A. (2011). Corporate governance in Bangladesh: A quest for the accountability or legitimacy crisis? 63. In S. Susela Devi & K. Hooper (Eds.), Accounting in Asia (Vol. 11, pp. 1-34). Emerald Group Publishing Limited. https://doi.org/10.1108/S1479-3563(2011)0000011006

VIRTUS 203

- 64. Saaydah, M. I. (2022). Introducing e-Disclosure (via XBRL) in Jordan, the benefits, cost and level of actual compliance. In A. Hamdan, H. M. Shoaib, B. Alareeni, & R. Hamdan (Eds.), *The implementation of smart technologies for business success and sustainability: During COVID-19 crises in developing countries* (pp. 247–261). Springer. https://doi.org/10.1007/978-3-031-10212-7_22
- 65. Salameh, R. S., Al-Nsour, O. J., Lutfi, K. M., Alnabulsi, Z. H., & Hyasat, E. A. H. (2023). The effect of board composition and ownership structure on firm value: Evidence from Jordan. *The Journal of Asian Finance, Economics and Business, 10*(2), 163–174. https://doi.org/10.13106/jafeb.2023.vol10.no2.0163
- 66. Scott, W. R. (2005). *Institutional theory: Contributing to a theoretical research program*. https://www.researchgate.net/publication/265348080_Institutional_Theory_Contributing_to_a_Theoretical_Res earch_Program
- 67. Securities Law of 2002. (2002). The Securities Commission. https://www.ase.com.jo/en/Legislation/Laws /Securities-Law
- 68. Selznick, P. (1996). Institutionalism "old" and "new." Administrative Science Quarterly, 41(2), 270-277. https://doi.org/10.2307/2393719
- 69. Shanikat, M., & Abbadi, S. S. (2011). Assessment of corporate governance in Jordan: An empirical study. *Australasian Accounting, Business and Finance Journal, 5*(3), 93–106. https://ro.uow.edu.au/aabfj/vol5/iss3/7/
- 70. Shbeilat, M., & Abdel-Qader, W. (2018). Independence dilemma and the reliability of the audit report: Qualitative evidence from Jordan. *Australian Academy of Accounting & Finanace Review*, 4(1), 22–36. https://www.researchgate.net/publication/328886489_Independence_Dilemma_and_the_Reliability_of_the_Audit _Report_Qualitative_Evidence_from_Jordan_Australian_Academy_of_Accounting_and_Finance_Review_AAAFR
- 71. Shen, L., Govindan, K., & Shankar, M. (2015). Evaluation of barriers of corporate social responsibility using an analytical hierarchy process under a fuzzy environment A textile case. *Sustainability*, *7*(3), 3493–3514. https://doi.org/10.3390/su7033493
- 72. Siddiqui, J. (2010). Development of corporate governance regulations: The case of an emerging economy. *Journal of Business Ethics*, *91*, 253–274. https://doi.org/10.1007/s10551-009-0082-4
- 73. Solas, C. (1994). Financial reporting practice in Jordan: An empirical test. *Advances in International Accounting*, 7(1), 43–60.
- 74. Steinfeld, J. M. (2023). Stewardship theory over agency theory. In J. M. Steinfeld (Ed.), *Public-private stewardship: Achieving value-for-money in public-private partnerships* (pp. 123–134). Springer. https://doi.org/10.1007/978-3-031-17131-4_8
- 76. Tsamenyi, M., Cullen, J., & González, J. M. G. (2006). Changes in accounting and financial information system in a Spanish electricity company: A new institutional theory analysis. *Management Accounting Research*, *17*(4), 409–432. https://doi.org/10.1016/j.mar.2006.02.002
- 77. Uddin, S., & Choudhury, J. (2008). Rationality, traditionalism and the state of corporate governance mechanisms: Illustrations from a less-developed country. *Accounting, Auditing & Accountability Journal, 21*(7), 1026–1051. https://doi.org/10.1108/09513570810907465
- 78. World Bank. (2021). Jordan's economic update April 2021. https://www.worldbank.org/en/country/jordan /publication/economic-update-april-2021
- 79. World Bank. (2023). *Jordan MPO*. https://thedocs.worldbank.org/en/doc/65cf93926fdb3ea23b72f277fc249a72-0500042021/related/mpo-jor.pdf
- 80. World Bank. (2023, March 23). *Governance and institutions*. https://ida.worldbank.org/theme/governance-and -institutions
- 81. Zattoni, A., & Cuomo, F. (2008). Why adopt codes of good governance? A comparison of institutional and efficiency perspectives. *Corporate Governance: An International Review*, *16*(1), 1–15. https://doi.org/10.1111/j .1467-8683.2008.00661.x
- 82. Zureigat, B. N., Fadzil, F. H., & Ismail, S. S. S. (2014). The relationship between corporate governance mechanisms and going concern evaluation: Evidence from firms listed on Amman Stock Exchange. *Journal of Public Administration and Governance*, *4*(4), 100–110. http://doi.org/10.5296/jpag.v4i4.6745

VIRTUS 204